Author's response to reviews

Title: The association between interpregnancy interval and birth weight: what is the role of maternal polyunsaturated fatty acid status?

Authors:

Luc JM Smits (luc.smits@epid.unimaas.nl)
Hester M Elzenga (h.elzenga@alumni.maastrichtuniversity.nl)
Reinoud JBJ Gemke (RJBGemke@vumc.nl)
Gerard Hornstra (G.Hornstra@Nutrisearch.NL)
Manon van Eijsden (MvEijsden@ggd.amsterdam.nl)

Version: 5 Date: 20 September 2012

Author's response to reviews:

Please find the final version of our paper. Alterations: 1) we changed phrasing as suggested by the reviewers; 2) we hired a professional translator to review and correct English; 3) we put the references in the right format.

We hope that the paper is now apt for publication in BMC Pregnancy and Childbirth.

Thank you
On behalf of the authors,

Luc Smits